\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage[authoryear]{natbib}



\newif\ifnokeywords\nokeywordsfalse

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}
\let\@issuelogo\@empty%
\let\@issuelogoex\@empty%
\def\issuelogo#1{\gdef\@issuelogo{#1}}
\def\issuelogoex#1{\gdef\@issuelogoex{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}
\lefthyphenmin = 3

\def\title#1{%
  \gdef\@title{%
      \vspace*{-40pt}%
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%%{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}\hfill
         \mbox{~}\hspace{5pt}\parbox{.1\linewidth}{\IfFileExists{images/\@issuelogo.\@issuelogoex}{\includegraphics[width=79pt,height=61pt]{images/\@issuelogo.\@issuelogoex}}{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}}\hfill
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
          \begin{center}\fontsize{17}{17}\selectfont\scshape\vskip-7pt International Journal of Research in Pharmaceutical Sciences\hfill\end{center}%
          \vspace*{-10pt}\hspace*{4pt}{\fontsize{8}{9}\selectfont Published by JK Welfare \& Pharmascope Foundation\hfill  Journal Home Page: \href{www.ijrps.com}{\color{blue}\underline{\smash{www.ijrps.com}}}}\hspace*{4pt}\mbox{}}}%
       \par\vspace*{-1pt}\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}

\def\abbrvJournalTitle{Int. J. Res. Pharm. Sci.}

\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}

                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
%\def\bibjtitle#1{#1}   %%Publisher request
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}

\makeatother
\date{}


\usepackage{float}

\begin{document}


\def\RunningAuthor{Lekha and Hannah R}
\firstPage{641}
\articleType{Original Article}
\receivedDate{30 Jun 2020}
\acceptedDate{31 Jul 2020 }
\revisedDate{20 Jul 2020}
\journalVolume{2020, 11 (SPL)}
\journalIssue{1}
\journalDoi{ijrps.v11iSPL1.3060}
\copyrightYear{2020}
\def\authorCount{2}
\def\affCount{1}

\def\journalTitle{International Journal of Research in Pharmaceutical Sciences}

\title{Evaluation of anxiety levels amidst Covid-19 pandemic among Chennai population}
\author{Lekha, 
            Hannah~R\textsuperscript{*}~\\[5pt]{Department of Oral and Maxillofacial Pathology and Microbiology, Saveetha Dental College and Hospital, Saveetha Institution of Medical and Technical Sciences, Saveetha University, Chennai {\textendash}600077, Tamil Nadu, India}}

\begin{abstract}
Anxiety refers to Intense, excessive and persistent worry and fear about everyday situations. Fast heart rate, rapid breathing, sweating and feeling of tiredness is how WHO defines anxiety. COVID 19 pandemic has increased the global anxiety level. The relatives and acquaintances infected with COVID - 19 is a risk factor for increasing the anxiety level. Self-administered questionnaires were designed based on knowledge attitude, and practice. The participants were in the age group of 18 - 60 years and belonged to the Chennai population. The questions were validated and distributed using google forms. After receiving enough responses, the data was collected and statistically analysed. The knowledge and awareness of COVID 19 are high in the Chennai population. Along with an increase in the anxiety level in the context of COVID 19. The results show the need for social support to alleviate stress and improve mental health
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}Anxiety level,\newline COVID 19,\newline Chennai population,\newline stress
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright\textsuperscript{*}Corresponding Author\par Name:\ Hannah~R~\\ Phone:\ 9962071806~\\  Email:\  hannahr.sdc@saveetha.com
        \par}\vspace*{-11pt}\hrulefill\par{\fontsize{12}{14}\selectfont ISSN: 0975-7538}\par% 
        \textsc{DOI:}\ \href{https://doi.org/10.26452/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/10.26452/\@journalDoi}}}}\par%
        \vspace*{-11pt}\hrulefill\\{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\par{IJRPS |~\href{www.ijrps.com}{www.ijrps.com}}\par%
        \vspace*{-2pt}{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\par%
        \vspace*{-11pt}\rule{\linewidth}{1.2pt}
        \makeatother
        
    
\section{Introduction}
Anxiety can be defined as an intense, excessive and persistent worry and fear about everyday situations. As per WHO, it can be accompanied with fast heart rate, rapid breathing, Sweating and feeling of tiredness. The significant categories of anxiety disorders are Panic disorder, agoraphobia without panic, social phobia, specific phobia, generalised anxiety disorder, acute stress disorder, post-traumatic stress disorder, Obsessive-compulsive disorder and anxiety disorder or not otherwise specified \unskip~\citep{857748:20065325}. The cognitive and behavioural approach has been widely used during the last two decades to precise the ecology of GAD \unskip~\citep{857748:20065318} Anxiety disorder side effects are decreased productivity. Increased mortality rates \unskip~\citep{857748:20065314}\ensuremath{\cdot} The COVID 19 pandemic increases global anxiety level \unskip~\citep{857748:20065311}. Mental state concerns of people impacted by the COVID 19 pandemic have not been adequately addressed. The objective of this study was to develop and evaluate the properties of the Coronavirus Anxiety Scale, which is a brief mental health screener to verify probable cases of anxiety associated with the coronavirus crisis. This study which was based on 775 adults with anxiety over the COVID 19, demonstrated solid reliability and validity. Elevated Coronavirus Anxiety Scale scores were found to be associated with coronavirus diagnosis, impairment, alcohol/drug coping, negative religious coping, extreme hopelessness, suicidal ideation, as well as Chinese products \unskip~\citep{857748:20065307}.  Many people suffer from uncertainty a fear of infection, moral distress and grief often experienced alone \unskip~\citep{857748:20065332}\ensuremath{\cdot} Since coronavirus is a newly emerging disease, more work is needed to find out the strategies to diagnose, prevent and treat the Covid 19 Pandemic. The current pandemic situation of Covid 19 is still severe \unskip~\citep{857748:20065320}. The relatives and acquaintances infected with Covid - 19 are a risk factor for increasing the anxiety levels \unskip~\citep{857748:20065316}.


\bgroup
\fixFloatSize{images/2aa13c00-9357-4cb1-8558-663e42e3cbab-upicture1.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/2aa13c00-9357-4cb1-8558-663e42e3cbab-upicture1.png}{\includegraphics{images/2aa13c00-9357-4cb1-8558-663e42e3cbab-upicture1.png}}{}
\makeatother 
\caption{\boldmath {pie chart depicting the age of the participants 65\% of people are female [blue zone] 35\% male [green zone]}}
\label{f-a5a9d6fe42aa}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/72473b9c-f260-475c-979c-68ec828abb2b-upicture2.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/72473b9c-f260-475c-979c-68ec828abb2b-upicture2.png}{\includegraphics{images/72473b9c-f260-475c-979c-68ec828abb2b-upicture2.png}}{}
\makeatother 
\caption{\boldmath {Pie chart depicting the response of the participants regarding awareness about COVID 19 }}
\label{f-023fca618cf3}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/d3a5a069-735a-49d7-8bb6-d62a95e1b059-upicture3.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/d3a5a069-735a-49d7-8bb6-d62a95e1b059-upicture3.png}{\includegraphics{images/d3a5a069-735a-49d7-8bb6-d62a95e1b059-upicture3.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts the association between age and their awareness about COVID 19}}
\label{f-8090bed5608c}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/39c9689f-fb67-48bd-9b16-cab44411b000-upicture4.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/39c9689f-fb67-48bd-9b16-cab44411b000-upicture4.png}{\includegraphics{images/39c9689f-fb67-48bd-9b16-cab44411b000-upicture4.png}}{}
\makeatother 
\caption{\boldmath {Pie chart depicting the response of the participants regarding the measures taken to prevent COVID 19 }}
\label{f-13440766ba18}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/47744549-8d2f-4b7d-beb8-a11b0bd66265-upicture5.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/47744549-8d2f-4b7d-beb8-a11b0bd66265-upicture5.png}{\includegraphics{images/47744549-8d2f-4b7d-beb8-a11b0bd66265-upicture5.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts the association of knowledge regarding various measures taken to contain the virus with age}}
\label{f-c24dfca5d03d}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/97573040-667a-4a3f-9b2f-99b87965a7ac-upicture6.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/97573040-667a-4a3f-9b2f-99b87965a7ac-upicture6.png}{\includegraphics{images/97573040-667a-4a3f-9b2f-99b87965a7ac-upicture6.png}}{}
\makeatother 
\caption{\boldmath {Pie chart showing the opinion of the participants regarding increase in stress level since COVID 19}}
\label{f-b83faac14127}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/b20e0de7-a723-40d5-911d-149cfa7c8e1a-upicture7.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/b20e0de7-a723-40d5-911d-149cfa7c8e1a-upicture7.png}{\includegraphics{images/b20e0de7-a723-40d5-911d-149cfa7c8e1a-upicture7.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts the association of the response of the participants regarding the impact of media in increasing the stress level since COVID 19 with age}}
\label{f-373c34b4c298}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/5a499385-5d16-45f0-8137-7edc5a4f32fd-upicture8.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/5a499385-5d16-45f0-8137-7edc5a4f32fd-upicture8.png}{\includegraphics{images/5a499385-5d16-45f0-8137-7edc5a4f32fd-upicture8.png}}{}
\makeatother 
\caption{\boldmath {Pie Chart indicating the form of media which most influences the stress levels of the participants }}
\label{f-3e54ce99fafd}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/17683b9b-4eb1-4863-ba77-1abbee833b95-upicture9.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/17683b9b-4eb1-4863-ba77-1abbee833b95-upicture9.png}{\includegraphics{images/17683b9b-4eb1-4863-ba77-1abbee833b95-upicture9.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts the association of responses on the form of media which influences their stress levels the most with age}}
\label{f-81e5d7a6de6c}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/f18df7c3-df2b-45b6-8142-0bfeae3eb603-upicture10.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/f18df7c3-df2b-45b6-8142-0bfeae3eb603-upicture10.png}{\includegraphics{images/f18df7c3-df2b-45b6-8142-0bfeae3eb603-upicture10.png}}{}
\makeatother 
\caption{\boldmath {Pie chart showing the opinion of the participants if they have visited a psychologist regarding anxiety issues due to COVID 19}}
\label{f-dad93d1612cf}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/241c282a-98a2-40a7-ad1e-ec3b26047a47-upicture11.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/241c282a-98a2-40a7-ad1e-ec3b26047a47-upicture11.png}{\includegraphics{images/241c282a-98a2-40a7-ad1e-ec3b26047a47-upicture11.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts association of response on the need to visit a psychologist regarding their anxiety issues due to COVID 19 with age}}
\label{f-123353bdc28c}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/735be72a-306a-4767-8245-25e4a70c7e25-upicture12.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/735be72a-306a-4767-8245-25e4a70c7e25-upicture12.png}{\includegraphics{images/735be72a-306a-4767-8245-25e4a70c7e25-upicture12.png}}{}
\makeatother 
\caption{\boldmath {Pie chart showing the opinion of the participants if they very anxious to stepping out of the house when you need essentials }}
\label{f-d2bd549e341a}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/92a15747-3960-4df5-b124-479f07d84d4f-upicture13.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/92a15747-3960-4df5-b124-479f07d84d4f-upicture13.png}{\includegraphics{images/92a15747-3960-4df5-b124-479f07d84d4f-upicture13.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts the association of responses regarding anxiety level of the participants when they step out of the house for essentials with age}}
\label{f-cec4a63dce07}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/4f36a29b-1ca2-4b21-8434-f7a4105d2047-upicture14.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/4f36a29b-1ca2-4b21-8434-f7a4105d2047-upicture14.png}{\includegraphics{images/4f36a29b-1ca2-4b21-8434-f7a4105d2047-upicture14.png}}{}
\makeatother 
\caption{\boldmath {Pie chart showing the opinion of the participants if have thought of COVID 19 which have made you feeldizzy, light headed, shaky or wooby}}
\label{f-c1265a57964a}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/6f6a72e2-88a9-42bd-8352-031db44f498c-upicture15.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/6f6a72e2-88a9-42bd-8352-031db44f498c-upicture15.png}{\includegraphics{images/6f6a72e2-88a9-42bd-8352-031db44f498c-upicture15.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts the association of the response on whether the thought of COVID19 makes them feel dizzy, light headed, shaky or wobbly with age group}}
\label{f-903ec4007551}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/fad08e52-91ed-4976-9bcd-47440f454d03-upicture16.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/fad08e52-91ed-4976-9bcd-47440f454d03-upicture16.png}{\includegraphics{images/fad08e52-91ed-4976-9bcd-47440f454d03-upicture16.png}}{}
\makeatother 
\caption{\boldmath {Pie chart showing the opinion of the participants if they have times when you feel like your beat heart is racing and you are hot and Sweaty thinking of COVID 19}}
\label{f-5b3f57f9a8e4}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/7d10ecc7-2bb0-4090-8dd9-32c4ed6de782-upicture17.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/7d10ecc7-2bb0-4090-8dd9-32c4ed6de782-upicture17.png}{\includegraphics{images/7d10ecc7-2bb0-4090-8dd9-32c4ed6de782-upicture17.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts the association of increased anxiety levels causing palpitations with age}}
\label{f-e29b6a510c58}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/d8cddc36-8e2a-4f51-9070-bd30e7637e31-upicture18.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/d8cddc36-8e2a-4f51-9070-bd30e7637e31-upicture18.png}{\includegraphics{images/d8cddc36-8e2a-4f51-9070-bd30e7637e31-upicture18.png}}{}
\makeatother 
\caption{\boldmath {Pie chart showing the opinion of the participants if their anxiety level increased since COVID 19 }}
\label{f-defae30ee34b}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/ee1d4403-e204-4135-b79c-c3195614d3c0-upicture19.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/ee1d4403-e204-4135-b79c-c3195614d3c0-upicture19.png}{\includegraphics{images/ee1d4403-e204-4135-b79c-c3195614d3c0-upicture19.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts the association of increase in anxiety level since COVID 19 with age}}
\label{f-ead55653ff71}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/c954d564-cc1e-45a5-a720-b6fec8a17717-upicture20.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/c954d564-cc1e-45a5-a720-b6fec8a17717-upicture20.png}{\includegraphics{images/c954d564-cc1e-45a5-a720-b6fec8a17717-upicture20.png}}{}
\makeatother 
\caption{\boldmath {Pie chart showing the opinion of the participants regarding the stress level}}
\label{f-86f0f2fec772}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/75423fea-8a0c-4090-acc9-76f5d998cc99-upicture21.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/75423fea-8a0c-4090-acc9-76f5d998cc99-upicture21.png}{\includegraphics{images/75423fea-8a0c-4090-acc9-76f5d998cc99-upicture21.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts the association of decreased stress level since lockdown with age group }}
\label{f-9e2da903faeb}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/20379d83-26b5-4b3f-9070-c65e8b06e402-upicture22.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/20379d83-26b5-4b3f-9070-c65e8b06e402-upicture22.png}{\includegraphics{images/20379d83-26b5-4b3f-9070-c65e8b06e402-upicture22.png}}{}
\makeatother 
\caption{\boldmath {Pie chart indicating the emotion of the participants when they hear the words Quarantine, Lockdown, Social distancing}}
\label{f-8b106c902991}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/cd2441de-3e94-4886-ba81-f65a47779fa1-upicture23.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/cd2441de-3e94-4886-ba81-f65a47779fa1-upicture23.png}{\includegraphics{images/cd2441de-3e94-4886-ba81-f65a47779fa1-upicture23.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts the association of emotions regarding Quarantine , lockdown,social distancing with age group}}
\label{f-99af35715adf}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/d49850aa-6913-4f14-92bd-57753e3ec962-upicture24.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/d49850aa-6913-4f14-92bd-57753e3ec962-upicture24.png}{\includegraphics{images/d49850aa-6913-4f14-92bd-57753e3ec962-upicture24.png}}{}
\makeatother 
\caption{\boldmath {Pie chart showing whether the participants are apprehensive or paranoid if someone coughs a few times at home}}
\label{f-bd3c2619805a}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/9eaf86e6-c2d3-4e24-830d-ef23b0973657-upicture25.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/9eaf86e6-c2d3-4e24-830d-ef23b0973657-upicture25.png}{\includegraphics{images/9eaf86e6-c2d3-4e24-830d-ef23b0973657-upicture25.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts the association of level of apprehensiveness when someone coughs a few times at home with age}}
\label{f-3390711167be}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/2f65e750-4a3d-44b7-a24a-2165d0819258-upicture26.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/2f65e750-4a3d-44b7-a24a-2165d0819258-upicture26.png}{\includegraphics{images/2f65e750-4a3d-44b7-a24a-2165d0819258-upicture26.png}}{}
\makeatother 
\caption{\boldmath {Pie chart indicating whether the participants fear that the worst will happen to them or their family members during this pandemic}}
\label{f-cccce2b6121e}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/4ef48ed8-5240-4ff4-9a12-307653a0031d-upicture27.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/4ef48ed8-5240-4ff4-9a12-307653a0031d-upicture27.png}{\includegraphics{images/4ef48ed8-5240-4ff4-9a12-307653a0031d-upicture27.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts the association of the response for fear that the worst will happen to them or their family members during this pandemic with age groups}}
\label{f-6c83d9bb6b91}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/51931b0b-724b-4bde-abfe-4460e11caf09-upicture28.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/51931b0b-724b-4bde-abfe-4460e11caf09-upicture28.png}{\includegraphics{images/51931b0b-724b-4bde-abfe-4460e11caf09-upicture28.png}}{}
\makeatother 
\caption{\boldmath {Pie chart indicating whether the participants suffer from lack of sleep due to the COVID 19 pandemic }}
\label{f-e1859347c6f5}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/7ee9ce4c-a8d0-4fa3-9504-1b9cae725ce4-upicture29.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/7ee9ce4c-a8d0-4fa3-9504-1b9cae725ce4-upicture29.png}{\includegraphics{images/7ee9ce4c-a8d0-4fa3-9504-1b9cae725ce4-upicture29.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts association of lack of sleep due to the COVID 19 pandemic with age groups}}
\label{f-7c2b46fdf82d}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/d1201819-0cd8-44fa-a912-63258ee5573f-upicture30.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/d1201819-0cd8-44fa-a912-63258ee5573f-upicture30.png}{\includegraphics{images/d1201819-0cd8-44fa-a912-63258ee5573f-upicture30.png}}{}
\makeatother 
\caption{\boldmath {Pie chart indicating whether the participants suffer from nightmares due to COVID19 pandemic}}
\label{f-f10a6b37edc7}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/2e601776-2050-4bea-896e-933f9856eb21-upicture31.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/2e601776-2050-4bea-896e-933f9856eb21-upicture31.png}{\includegraphics{images/2e601776-2050-4bea-896e-933f9856eb21-upicture31.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts the association of the responses on whether they suffer from nightmares due to COVID 19 pandemic with the age groups}}
\label{f-983a6c28af56}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/ca2b76c6-d806-4a9e-adea-1d9a44c81c1c-upicture32.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/ca2b76c6-d806-4a9e-adea-1d9a44c81c1c-upicture32.png}{\includegraphics{images/ca2b76c6-d806-4a9e-adea-1d9a44c81c1c-upicture32.png}}{}
\makeatother 
\caption{\boldmath {Pie chart showing whether the participants have been obsessed with news lately}}
\label{f-a9cb4d266238}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/b9817c02-d1a0-4e2a-9766-ee5250938741-upicture33.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/b9817c02-d1a0-4e2a-9766-ee5250938741-upicture33.png}{\includegraphics{images/b9817c02-d1a0-4e2a-9766-ee5250938741-upicture33.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts the association of the obsession with COVID related news with age groups}}
\label{f-da7fb4ed8b3e}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/8f9ef803-1bd1-459f-98fd-17f2bc34faa2-upicture34.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/8f9ef803-1bd1-459f-98fd-17f2bc34faa2-upicture34.png}{\includegraphics{images/8f9ef803-1bd1-459f-98fd-17f2bc34faa2-upicture34.png}}{}
\makeatother 
\caption{\boldmath {Pie chart indicating whether the participants get anxious on listening to news regarding COVID 19 }}
\label{f-60b0a7973aa2}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/7217c062-9570-45c5-b83b-c861860697d2-upicture35.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/7217c062-9570-45c5-b83b-c861860697d2-upicture35.png}{\includegraphics{images/7217c062-9570-45c5-b83b-c861860697d2-upicture35.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts the association of opinionon whether news regarding COVID 19 causes anxiety with age groups}}
\label{f-ef09b9d41363}
\end{figure}
\egroup
In Study by Huang Y et al. about anxiety levels during COVID 19 in China explained about anxiety disorder, depressive symptoms and sleep quality, the author identified a significant mental health burden in the Chinese public during COVID 19 outbreak \unskip~\citep{857748:20065311}. Young people were spending too much time thinking about the outbreak, and the healthcare workers were at high risk of displaying psychological issues \unskip~\citep{857748:20065311}. Although sources of anxiety may not have affected everyone, they can weaken the confidence of health care workers in themselves. The health care delivery source says their ability to stay calm and assure the public is most needed. According to the sources, anxiety allows health care workers and organisations to develop targeted approaches to address these queries and provide specific support to their health care workers \unskip~\citep{857748:20065329}. About 80\% of the population feels anxious in the Current Situation \unskip~\citep{857748:20065321}. In the Study by Stein MB, they found out that there is increased anxiety and depression among people thinking about their future with covid  19 \unskip~\citep{857748:20065333}. A nationwide survey of more than 50,000 people in China during Covid 19 epidemic showed that 35\% of the respondents experience psychological distress \unskip~\citep{857748:20065317}.


\bgroup
\fixFloatSize{images/bcd7c05e-bea5-4405-9a04-6677b3f4c94c-upicture36.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/bcd7c05e-bea5-4405-9a04-6677b3f4c94c-upicture36.png}{\includegraphics{images/bcd7c05e-bea5-4405-9a04-6677b3f4c94c-upicture36.png}}{}
\makeatother 
\caption{\boldmath {Bar Graph showing the anxiety level of the participants in a scale from 1 to 10}}
\label{f-629cf7bb1990}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/b710c53b-ed56-4876-bdb9-37a2cf142d85-upicture37.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/b710c53b-ed56-4876-bdb9-37a2cf142d85-upicture37.png}{\includegraphics{images/b710c53b-ed56-4876-bdb9-37a2cf142d85-upicture37.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts association of anxiety level of the participants in a scale from1 to 10 with age groups}}
\label{f-a0844d128ecf}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/1388a784-89b1-415c-ba9f-4d43447fa1a5-upicture38.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/1388a784-89b1-415c-ba9f-4d43447fa1a5-upicture38.png}{\includegraphics{images/1388a784-89b1-415c-ba9f-4d43447fa1a5-upicture38.png}}{}
\makeatother 
\caption{\boldmath {Pie chart showing whether the participants take medication in order to bring down their anxiety level }}
\label{f-9405758c66ba}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/f461e2e4-a3a4-44e6-bf64-983db2c1942d-upicture39.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/f461e2e4-a3a4-44e6-bf64-983db2c1942d-upicture39.png}{\includegraphics{images/f461e2e4-a3a4-44e6-bf64-983db2c1942d-upicture39.png}}{}
\makeatother 
\caption{\boldmath {Bar graph depicts the association of the participants who take medication in order to bring down their anxiety level with age groups}}
\label{f-bb6bb4aef52f}
\end{figure}
\egroup
The effects were mostly related to the thoughts of loneliness, anxiety and post-traumatic stress disorder symptoms \unskip~\citep{857748:20065308}. It should be pointed out that the nature of the pandemic has led to constant updating of the media or television on COVID 19, which may increase the level of anxiety. Excessive consumption of knowledge about the spread of communicating diseases may lead to persistent thinking and can affect the mental health of individuals \unskip~\citep{857748:20065308}. The ability to bring pandemics under control depends on compliance with a warning. One psychological factor capable of affecting adherence to the sign is health anxiety \unskip~\citep{857748:20065315}. More than 203 countries, areas or territories have been affected by the COVID 19 virus still now, And about 630,000 infected and nearly 30,000 deaths reported by March 29. Iran, after Italy, Spain and China, has had the most significant number of deaths in the entire world. Evidence also shows that individuals may experience symptoms of, anxiety, trauma, suicidal ideation, and panic during outbreaks of infectious diseases \unskip~\citep{857748:20065310}. 

Many institution-based studies have been carried out on various oral diseases \unskip~\citep{857748:20065330,857748:20065323,857748:20065322,857748:20065304,857748:20065324}. There have also been studies done using recent technology to enhance knowledge \unskip~\citep{857748:20065327,857748:20065313,857748:20065326,857748:20065319,857748:20065331}. Many surveys were also conducted to improve the level of awareness regarding various diseases and recent development among the general population and the dental students \unskip~\citep{857748:20065309,857748:20065312,857748:20065306,857748:20065305,857748:20065328}. The current study was conducted in the light of the studies mentioned above to evaluate the knowledge, attitude and practice regarding anxiety level during Covid 19 pandemic among Chennai population.

\textbf{ Methodology}

This is a cross-sectional descriptive Survey that is conducted among 108 people in the age group between 20-60 years. This survey consists of 18 self-administered questions. This was done through an online survey (google forms) This survey was carried among Chennai population that consists of both male and female between the age between 20-60 Years. Chi-square analysis to compare the two groups is done using SPSS. Simple Random sampling was done to eliminate response bias. Demographic Information knowledge, Attitude and Practice were the output of variable. Age, gender, physical problems, work were the independent Variables, whereas knowledge, Altitude and Practice were the dependent variables. The results were collected and were tabulated. The results were analysed using statistical analysis.
    
\section{Results and Discussion}
The descriptive data showed that the majority of the population that actively participated in the survey were 18 years of age. In this study, the male population was 34.7\%, whereas the female population was 65.3\%. The participants engaged in this survey, 44\% were students. 5\% of the population were retired .3\%of the population were teachers.2\% of the population were lawyers. 9\% of the population were health workers .7\% of the population were Entrepreneur.17\% of the population were engineers.10\% of the population were homemakers. 2\% of the population were scientists.1\% of the population were farmers. 

On assessing the knowledge of the participants, 94\% of the participants were aware of COVID 19 pandemic and on comparing between the {\textless}30 and {\textgreater}30 age groups, statistically significant increase in awareness was seen among {\textless}30 age group Figures~\ref{f-a5a9d6fe42aa}, \ref{f-023fca618cf3} and~\ref{f-8090bed5608c} shows that Chi square test shows statistical significance (P = 0.009) showing more awareness among {\textless}30 years of age.When the participants were asked what measures were taken to contain the spread of virus 6\% of the population said Quarantine 15\% said lockdown 16\% said Social distancing and 63\% said all of the above, on comparing between the groups no difference was found Figures~\ref{f-13440766ba18} and~\ref{f-c24dfca5d03d} shows that Chi square analysis shows no statistical significance (P=0.721). When the participants were asked if the media increased their anxiety level, 56\% of the population said yes, and 44\% said no. Still, no difference was seen between the groups Figures~\ref{f-b83faac14127} and~\ref{f-373c34b4c298} sows that Chi square test shows no statistical significance with P=0.658 (P{\textgreater}0.05). When the participants were asked which form of media had the most influence in their stress levels, 42\% of the population said television, 10\% said Newspapers, 26 \% said social media and 21\% said  No influence of media. Still, no difference in response was seen between the groups Figures~\ref{f-3e54ce99fafd} and~\ref{f-81e5d7a6de6c} shows that Chi square test shows no statistical significance with P = 0.056 (P{\textgreater}0.05).

On assessing the attitude of the participants, 78\% of the participants felt the need to visit a professional psychologist to help with anxiety issues. Still, Chi-square test did not show a statistically significant difference between the groups Figures~\ref{f-dad93d1612cf} and~\ref{f-123353bdc28c} shows that Chi square test shows no statistical significance with P =0.781(P{\textgreater}0.05). When the participants were asked if were anxious to step out of the house for essentials, 50\% of the population answered yes, but Chi-square test did not show a statistically significant difference between the groups Figures~\ref{f-d2bd549e341a} and~\ref{f-cec4a63dce07} shows that Chi square test shows no statistical significance with P=0.091(P{\textgreater}0.05). When the participants were asked whether the thought of COVID 19 made you feel dizzy, light-headed, shaky or wobbly 25\% of the population answered yes, but no significant difference was seen between the groups Figures~\ref{f-c1265a57964a} and~\ref{f-903ec4007551} shows that Chi square test shows no statistical significance with P = 0.88(P{\textgreater}0.05). When the participants were asked if they have had times when they feel like their heartbeat is racing, and they are hot and Sweaty thinking of COVID 19, 34\% of the participants answered yes. Still, no significant difference was seen between the groups Figures~\ref{f-5b3f57f9a8e4} and~\ref{f-e29b6a510c58} shows that Chi square test shows no statistical significance P = 0.803(P{\textgreater}0.05). When the participants were asked if their anxiety level has increased since COVID 19, 65\% of the population answered yes, Chi-square test showed statistical significance with more increase in anxiety level among the {\textless}30 age group Figures~\ref{f-defae30ee34b} and~\ref{f-ead55653ff71} shows that Chi square analysis shows statistical significance P = 0.024 (P{\textless}0.05). When the participants were asked if lockdown decreased their stress level, 50\% of the population said yes, but no significant difference was seen between the groups Figures~\ref{f-86f0f2fec772} and~\ref{f-9e2da903faeb} shows that Chi square test shows no statistical significance with P =0.205(P{\textgreater}0.05). When the participants were asked about their emotion when they hear the words quarantine, lockdown and social distancing.  42\% of the population answered sadly, 22 \% said happy, 13\% said anger and 23\% said fear, but no significant difference was seen between the groups Figures~\ref{f-8b106c902991} and~\ref{f-99af35715adf} shows that Chi square test shows no statistical significance with P =0.539(P{\textgreater}0.05). When the participants were asked if they were apprehensive or paranoid if someone coughs a few times at home, 25\% of the population answered yes, and 29\% sometimes answered Figure~\ref{f-bd3c2619805a}. Chi-square analysis shows statistical significance P = 0.012. More anxiety is seen among \textless\ 30 age group Figure~\ref{f-3390711167be} shows that Chi square analysis shows statistical significance P = 0.012. More anxiety is seen among \textless\ 30 age group. When the participants were asked if they fear that the worst will happen to their family members during this pandemic, 25\% of the population answered yes Figures~\ref{f-cccce2b6121e} and~\ref{f-6c83d9bb6b91} shows that Chi square analysis shows statistical significance(p value 0.012). More fear is seen among {\textless}30 age group. Chi-square analysis showed statistical significance (p-value 0.012) with more fear seen among {\textless}30 age group.

On assessing the Practice, When the participants were asked if they suffered from lack of sleep thinking about COVID 19, 15\% of the population said yes, and 64\% answered no. Still, no significant difference was seen between the groups Figures~\ref{f-e1859347c6f5} and~\ref{f-7c2b46fdf82d} shows that Chi square test shows no statistical significance with P = 0.611(P{\textgreater}0.05). When the participants were asked if they get nightmares about COVID 19, 24\% of the population answered yes, but no significant difference was seen between the groups Figures~\ref{f-f10a6b37edc7} and~\ref{f-983a6c28af56} shows that Chi square test shows no statistical significance with P = 0.058(P{\textgreater}0.05). When the participants were asked if they were obsessed with news regarding COVID 19, 61\% of the population answered yes Figure~\ref{f-a9cb4d266238}. Chi-square analysis showed statistical significance with more obsession among \textless\ 30 age group Figure~\ref{f-da7fb4ed8b3e} shows that Chi square analysis shows statistical significance P = 0.049 (P{\textless}0.05), More obsession is seen among\textless\ 30 age group. When the participants were asked if listening to news regarding COVID 19 cause anxiety, 61\% of the population said yes, but no significant difference was seen between the groups Figures~\ref{f-60b0a7973aa2} and~\ref{f-ef09b9d41363} shows that Chi square analysis shows no statistical significance with P = 0.236 (P{\textgreater}0.05). When the participants were asked On the scale of 1-10 how anxious they were about COVID 19, 17\% of the population said 1, 5\% said 2,5\% said 3, 5\% said 4,16\% said5, 9\% said 6,10\% said 7, 13\% said 8,6\% said 9,16\% said10, but no significant difference was seen between the groups Figures~\ref{f-629cf7bb1990} and~\ref{f-a0844d128ecf} shows that Chi square analysis shows no statistical significance P =0.549 (P{\textgreater}0.05). When the participants were asked if here there were days when they had taken medication to bring their anxiety levels down,  19\% of the population answered yes. Still, no significant difference was seen between the groups Figures~\ref{f-9405758c66ba} and~\ref{f-bb6bb4aef52f} shows that Chi square analysis shows no statistical significance with P =  0.407(P{\textgreater}0.05).

In our study, 35\% of the participants in the Chennai population showed increased anxiety levels amidst COVID 19 pandemic. In contrast, in a study done by Roy D et al. the anxiety level on COVID 19 pandemic in Indian population is 80\%. Comparing to Roy D et al. study in our study, the anxiety level is a little lower. The difference may be because, in our research, predominant populations were lesser than 30 years of age \unskip~\citep{857748:20065321}. In a study done by Huang Y et al. amidst COVID 19 in the Chinese population, he evaluated anxiety disorder, depressive symptoms and sleep quality to be more prevalent. The anxiety level was 30\% which is similar to our study, where it is 35\%. The knowledge of the participants regarding COVID 19 was good which agreed with research done by Roy D et al. and Huang Y et al. This increased knowledge and awareness can be attributed to increased exposure to media by the participants \unskip~\citep{857748:20065311,857748:20065321}. COVID 19 is an infectious disease caused by a novel coronavirus, and it causes anxiety and mental stress to people.

Roy et al. in their study deciphered that for 36.5\% of the participant's social media was the primary form of media which influenced their stress levels which is similar to the results of our study where media influence 26\%. In today's world of living with COVID 19 pandemic, the role of the psychologist seems to be crucial. Roy D et al. in their study showed that 76\% of the participants needed to go to see a psychologist, whereas, in contrast, our study results show that only 33\% needed to go to the psychologist. The variations in the result may be because of less awareness about the role of psychologists among the Chennai population and also more awareness about the pandemic and coping mechanisms \unskip~\citep{857748:20065321}. In the study done by Ozdin, the majority of the population needed some psychological support. But in our research, only 33\% required support. This variation in the results may be because the study by Ozdin et was carried among people above 60 years of age which is in contrast with our study. Populations above the age group of 60 are known to succumb more to the disease and hence are naturally more apprehensive. So participants are more anxious and need the help of psychological Support \unskip~\citep{857748:20065315}. Psychologists are trained to diagnose anxiety disorder and teach patients an effective way to cope up. CBT therapy by psychologists is a very effective way to treat anxiety disorder

In the study done by Amir Moghani Bashi, the findings show that more than 94\% of people follow the news of COVID 19, but in our study, only 50\% of the population follow the news regarding COVID 19. The variation in results may be because the other research was done among the Iran population, and the awareness and social stigma towards COVID 19 may be different from the Indian population. But the anxiety level among the Iran population for COVID 19 is similar to our study, which is 33\% \unskip~\citep{857748:20065310}. Some of the measures taken by the participants to combat anxiety associated with covid19 are to take a time out, to eat well-balanced food, to have enough sleep, to pray and meditate. 

The limitation of our study is the less sample size, response bias and survey fatigue. It is also done in a homogenous population which may mislead.

The Future scope of this study is to do a survey-based complete survey on the Anxiety level of COVID 19 during the lockdown and its effects on mental health in a larger population and a larger scale.
    
\section{Conclusion}
This study has shown an increase in the anxiety level among the Chennai population in the context of COVID19 pandemic. The manifestation of anxiety level is more among {\textless}30 age group, and this is attributed to the increased exposure to information from media. The study reflects a need for more awareness programs about coping with anxiety levels during COVID19 among the Chennai population. There is a need for social support to alleviate anxiety and improve mental health. 

\textbf{Funding Support}

The authors declare that they have no funding support for this study.

\textbf{Conflict of Interest}

The authors declare that they have no conflict of interest for this study.
    

\bibliographystyle{pharmascope_apa-custom}

\bibliography{\jobname}
\end{document}
